Author's response to reviews

Title: Solitary breast metastasis from myxoid liposarcoma.

Authors:

Masahiro Yokouchi (masahiro@m3.kufm.kagoshima-u.ac.jp)
Satoshi Nagano (naga@m2.kufm.kagoshima-u.ac.jp)
Yuko Kijima (ykijima@m3.kufm.kagoshima-u.ac.jp)
Heiji Yoshinaka (heiji@m3.kufm.kagoshima-u.ac.jp)
Hirofumi Shimada (mac1mickmick2@yahoo.co.jp)
Shunsuke Nakamura (hhxt0y@hotmail.co.jp)
Takako Yoshioka (yoshioka@m2.kufm.kagoshima-u.ac.jp)
Akihide Tanimoto (akit09@m3.kufm.kagoshima-u.ac.jp)
Shoji Natsugoe (natsugoe@m2.kufm.kagoshima-u.ac.jp)
Setsuro Komiya (skomiya@m3.kufm.kagoshima-u.ac.jp)

Version: 3 Date: 26 May 2014

Author's response to reviews:

Dear Editor:

Thank you very much for your valuable comments on our manuscript. We have comprehensively modified our paper in accordance with the reviewer’s comments and requests, and wish to resubmit the revised manuscript. The revised portions in the manuscript are denoted in red font.

Our replies to the reviewers’ comments are listed below.

Reviewer 1

Following the reviewer’s comments on the case presentation, we have corrected the sentence as follows:

Original sentence, “No calcification was present.”
Revised sentence, “Macrocalcifications were not observed.”

In accordance with the reviewer’s comments about the discussion, we have added the following sentence:

“Previous reports suggested that breast metastasis from different primary tumors showed distinct radiological patterns [3, 11, 12]. Recently, Abbas et al. described that breast metastasis occurred more frequently in malignant melanomas, sarcomas, lung cancers, ovarian tumors, renal carcinomas, and thyroid tumors. In addition, they have described the mammographic and ultrasonographic features of these tumors [13]. Differentiating primary from metastatic breast
malignancies is important for rational management. In a patient with a known extramammary malignancy and a breast mass, mammograms and ultrasonograms are important to exclude primary breast carcinoma and for appropriate management [12, 13].”

In addition, we have cited the following references in the revised manuscript:


We would appreciate your evaluation of our revised manuscript and look forward to your response.

Respectfully yours,

Masahiro Yokouchi, MD, PhD
Department of Orthopaedic Surgery,
Kagoshima Graduate School of Medical and Dental Sciences,
8-35-1 Sakuragaoka, Kagoshima,
Japan 890 - 8520
Tel.: +81-99-275-5381
Fax: +81-99-265-4699
E-mail: masahiro@m3.kufm.kagoshima-u.ac.jp